Author’s response to reviews

Title: Dexamethasone for the prevention of a pain flare after palliative radiotherapy for painful bone metastases: a multicenter double-blind placebo-controlled randomized trial

Authors:

Paulien G Westhoff MD (p.g.westhoff@umcutrecht.nl)
Alexander de Graeff MD PhD (a.degraeff@umcutrecht.nl)
Jenske I Geerling (j.i.geerling@umcg.nl)
Anna K L Reyners MD PhD (a.k.l.reyners@umcg.nl)
Yvette M van der Linden MD PhD (y.m.van_der_linden@lumc.nl)

Version: 4 Date: 3 April 2014

Author’s response to reviews: see over
Dear sir/madam,

I hereby send you our manuscript entitled “Dexamethasone for the prevention of a pain flare after palliative radiotherapy for painful bone metastases: a multicenter double-blind placebo-controlled randomized trial”, to be considered for publication in “BMC Cancer”.

As you requested, we deleted the additional files with proof of ethics and funding from the revision. Secondly, we added the participating centers to the revised manuscript.

This manuscript describes the background and study design of our multicenter double-blind placebo-controlled randomized trial. The purpose of this trial is to study the effectiveness of dexamethasone to prevent a pain flare after palliative radiotherapy for painful bone metastases and to determine the optimal dose schedule. So far, 140 patients of the 411 patients required have been included. Since the risks of the study using dexamethasone are considered minimal, an interim analysis will not be performed. No randomized trials have been published that investigated the effect of dexamethasone in prevention of a pain flare after palliative radiotherapy for painful bone metastases.

The trial follows the CONSORT guidelines and is registered at ClinicalTrials.gov (NCT01669499). The trial receives funding from the Dutch Cancer Society (grant no. UU 2009-4595) and ZonMw, a Dutch organization for health research and innovation of healthcare (grant no.11510009). The study protocol was approved by the Medical Ethics Committees. Proof of funding and ethics were provided earlier.

All authors have read and approved the manuscript. On behalf of all authors, I state that there are no conflicts of interest and no financial disclosures. This manuscript is not under consideration elsewhere.

I hope you will consider this manuscript suitable for publication in your journal. If necessary, I can provide you with supplementary information.

Looking forward to your reply,

Yours sincerely,

Paulien Westhoff, M.D.